Author’s response to reviews

Title: Identification of compound heterozygous patients with Primary Hyperoxaluria Type 1: Clinical Evaluations and In Silico Investigations

Authors:

Houda Kanoun (houda.kanoun@gmail.com)
Faiçal Jarraya (jarraya_faical@yahoo.fr)
Bayen Maalej (mm_bayen@yahoo.fr)
Amina Lahiani (amina_lahyani@yahoo.fr)
Hichem Mahfoudh (hichem.mahfoudh@yahoo.fr)
Fatma Makni (ayadi-fatma@medecinesfax.org)
Jamil Hachicha (jamil.hachicha@rns.tn)
Faiza Fakhfakh (faiza.fakhfakh02@gmail.com)

Version: 2 Date: 23 Apr 2017

Author’s response to reviews:

Dear Editor,

Thank you very much for your assessment and decision letter on our revised manuscript entitled “Identification of compound heterozygous patients with Primary Hyperoxaluria Type 1: Clinical Evaluations and In Silico Investigations” (BNEP-D-16-00211R1). We have revised the manuscript according to the comments.

We hope that the revision is acceptable for publication in your journal.

Look forward to hearing from you soon.

Yours sincerely,

Kanoun Houda
1. Consent to Publish
You should have obtained consent to publish from the four participants.
Response:
Written informed consent was obtained from the four participants of the article. For patients P2 and P3 (under 18 years old), informed consent was obtained from their father.

2. Ethical Approval
It is not clear whether you obtained ethical approval for your study. You state in the Declarations that you obtained informed consent from patients in accordance with the ethical committee. Can you please revise this statement to clarify whether the ethical committee of Hedi Chaker Hospital approved the study protocol for this research.
Response:
The ethical committee of Hedi Chaker Hospital approved in principle the study protocol for this research. A written official agreement will be available on 6 May 2017, date of its next meeting.

3. Copyedit
I'm afraid there are still several spelling and grammatical errors throughout the manuscript.
Response:
For English Language improvements, our manuscript was reviewed by Pr Abdelmajid damak, founder and former Head of the English Unit at Sfax National school of engineer, Tunisia.
I hope that the quality of the written English manuscript is acceptable for publication.